Author’s response to reviews

Title: Mediterranean Diet Effect: An Italian picture

Authors:

Elena Azzini (azzini@inran.it)
Angela Polito (polito@inran.it)
Alessandro Fumagalli (fuma_a@alice.it)
Federica Intorre (intorre@inran.it)
Eugenia Venneria (venneria@inran.it)
Alessandra Durazzo (durazzo@inran.it)
Maria Zaccaria (zaccaria@inran.it)
Donatella Ciarapica (ciarapica@inran.it)
Maria S Foddaï (foddaï@inran.it)
Beatrice Mauro (mauro@inran.it)
Anna Raguzzini (raguzzini@inran.it)
Lara Palomba (palomba@inran.it)
Giuseppe Maiani (maiani@inran.it)

Version: 4 Date: 13 November 2011

Author’s response to reviews:

We checked and revised the most recent version of our manuscript:
we have listed all authors as suggested,
we have removed all highlighting and tracking from the manuscript
we have uploaded the figure separately from the manuscript, giving the title after
the reference list,
we have renamed all tab as table.
Thank you for your acceptance of the manuscript
The Authors